Author’s response to reviews

Title: Knowledge, Attitudes and Practices of grade three primary schoolchildren in relation to schistosomiasis, soil transmitted helminthiasis and malaria in Zimbabwe

Authors:

Nicholas Midzi (midzinicholas@yahoo.com)
Sekesai Mtapuri-Zinyowera (sekesaiz@yahoo.com)
Munyaradzi P Mapingure (mmapingure@yahoo.com)
Noah H Paul (npaul@science.uz.ac.zw)
Davison Sangweme (dsangweme@yahoo.com)
Gibson Hlerema (ghlerema@gmail.com)
Masceline J Mutsaka (mjmutsaka@yahoo.com)
Farisai Tongogara (ftongogara@gmail.com)
Godfrey Makware (gmakware@yahoo.com)
Vivian Chadukura (vchadukura@yahoo.com)
Kimberly C Brouwer (kbrouwer@ucsd.edu)
Francisca Mutapi (fmutapi@ed.ac.uk)
Nirbhay Kumar (nkumar@tulane.edu)
Takafira Mduluza (tmduluza@bhp.org.bw)

Version: 6 Date: 5 May 2011

Author’s response to reviews:

Revisions done on the version of the manuscript:

Author details revised and the author list amended as in the online submission specification.

Author (f) added as was missing from the submission

Corresponding authors contact details added.

Introduction heading changed to background

Conclusions added at the end of Discussions

Reference (20) was missing in the text this had been corrected and used appropriately.

Tables 5 and 6 that were mentioned in the text have been removed. This was an error that carried over from the old manuscript version.

The Figure legend has been added at the end of references

Tables have been included in the manuscript at the end of the reference list.

The Figure has been uploaded correctly without the legend.
Thank you.